**Author's response to reviews**

**Title:** Quality of life and wellbeing among HIV outpatients in East Africa: a multicentre observational study

**Authors:**

- Richard Harding (richard.harding@kcl.ac.uk)
- Victoria Simms (victoria.simms@lshtm.ac.uk)
- Suzanne Penfold (Suzanne.Penfold@lshtm.ac.uk)
- Julia Downing (Julia.Downing792@btinternet.com)
- Eve Namisango (eve.namisango@africanpalliativecare.org)
- Richard Powell (richard2powell@yahoo.co.uk)
- Faith Mwangi-Powell (mwangi1powell@yahoo.co.uk)
- Scott Moreland (smoreland@futuresgroup.com)
- Nancy Gikaara (nancy@kehpca.org)
- Mackuline Atieno (mackuline.atieno@africanpalliativecare.org)
- Irene J Higginson (irene.higginson@kcl.ac.uk)

**Version:** 3  
**Date:** 14 October 2014

**Author's response to reviews:** see over
Dear Editors

**MS: 1102446873138568**

Quality of life and wellbeing among HIV outpatients in East Africa: a multicentre observational study

Thank you for your email accepting our manuscript subject to these minor modifications, each addressed below.

<table>
<thead>
<tr>
<th>Comment</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Please clarify if it was written informed patient consent that was obtained, ensure that each section is labelled appropriately such as the 'Reference' and 'Conclusion' sections.</td>
<td>We have clarified this.</td>
</tr>
<tr>
<td>Please include contributions of Eve Namisango, Richard Powell and Faith Mwangi-Powell in 'Author's Contribution' since these are currently missing</td>
<td>We have clarified this.</td>
</tr>
</tbody>
</table>

Yours faithfully

Dr Richard Harding (corresponding author)